Author’s response to reviews

Title: Magnetic Resonance Tracking of Transplanted Autologous Bone Marrow-Derived Mesenchymal Stromal Cell in Chronic Spinal Cord Injury at Upper Cervical Level: Case Report

Authors:

Areesak Chotivichit (areesak.cho@mahidol.ac.th)
Monchai Rungchainikom (monchai.ortho@gmail.com)
Pipat Chiewvit (pipat8999@yahoo.com)
Adisak Wongkajornsilp (adisak.won@mahidol.ac.th)
Kittipong Suirattanawimol (Kittipong.suj@mahidol.ac.th)

Version: 2
Date: 3 September 2014

Author’s response to reviews: see over
Dear Stem Cells Research & Therapy

I revised the manuscript as your suggestion, I hope this version is following all of the instructions. The changing data in revised manuscript including

- Email addresses of all co-authors on the title page of the manuscript
- The manuscript title and author list to appear once
- The tables was placed at the end of the manuscript
- The figure legends was placed at the end of the manuscript

Regards,

Monchai Ruangchainikom, MD
Department of Orthopaedic Surgery
Faculty of Medicine Siriraj Hospital, Mahidol University
monchai.ortho@gmail.com